Author's response to reviews

Title: Experiences of undergoing cardiac surgery among older people diagnosed with postoperative delirium: one year follow-up

Authors:

Helena Claesson Lingehall (helena.lingehall@umu.se)
Nina Smulter (nina.smulter@umu.se)
Birgitta Olofsson (birgitta.olofsson@umu.se)
Elisabeth Lindahl (elisabeth.lindahl@umu.se)

Version: 4  Date: 13 March 2015

Author's response to reviews:

Umeå University
Department of Nursing
Umeå university, Sweden
SE- 901 87 Umeå SWEDEN
Helena Claesson Lingehall RN, PhD-student
MS: 3933540161542491 - revisions required prior to editorial assessment

Dear Editor

We apologize for the inconvenience. The point-by-point descriptions of the changes are made in manuscript. The abstract is now included in the main text.

Yours Sincerely

Helena Claesson Lingehall

Address for correspondence:

Helena Claesson Lingehall
RN, MSc, PhD-student, Department of Nursing, Umeå, Sweden,
Umeå University
SE- 901 87 Umeå SWEDEN
helena.lingehall@umu.se

Point-by-point description of the changes:

1. Kindly include the abstract in the main text.

The abstract is now included in the main text.